%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%% LaTeX Homework Template %%%%
%%%%%% Modified 17 Jan 2012   %%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\documentclass[12pt]{article}
\usepackage{amsfonts,amssymb,amsthm,amsmath}

\setlength{\oddsidemargin}{-0.1in} \setlength{\textwidth}{6.5in}
\setlength{\topmargin}{-.75in} \setlength{\textheight}{9.75in}

\newenvironment{problems}{\begin{list}{}{\setlength{\labelwidth}{.7in}}}{\end{list}}
\newcounter{problempart}
\newenvironment{parts}{\begin{list}{(\alph{problempart})}{\setlength{\itemsep}{0pt}\usecounter{problempart}}}{\end{list}}

\linespread{1.1} %Put some extra space between the lines.
\newcommand{\notR}{\not{\hskip -3pt R}}

\begin{document}


% This will be the standard header to put on all of your homework, so you
% only need to change it to your section number, Homework number, and your name

\noindent { \sc Math 290 Section 007 \hfill
Homework \# ?  \hfill %Title of the document
Your Name %Your Name Here
}

\bigskip

\begin{problems}


\item[ 3.1.2 ]%number of the problem





\end{problems}
\end{document}
